Dear Reviewer,

Thank you very much for your review and I would like to express my sincere gratitude to you. Based on the review comments, the following changes have been made to the article.

1. The sentence- “The endovascular approach represents an alternative with low mortality and morbidity rates and invasion.”- has changed as follows: The endovascular approach represents a minimally-invasive alternative with low mortality and morbidity.

2. The sentence in the introduction- “Obviously, our patient had reached the indications for surgical intervention.”- has deleted.

3. This patient was transferred to our department on the same day of admission due to CT findings of abdominal aortic coarctation and hepatic aneurysm. And CT indeed CT suggests abdominal aortic coarctation and hepatic aneurysm.

4. The description of the arrow color in Figure 1 has been changed to blue.

5. The term “violated” has changed to involving.

6. The sentence- “Embolization was also excluded because the HAA was so large that it might compress the biliary tract and duodenum and thus cause jaundice, gastrointestinal obstruction, and even duodenum fistula.” - was modified and retained only once.

7. The word “gender” in the sentence- “Lesions in the hepatic circulation show a ratio of approximately 3:2 in terms of gender with male predominance”- has changed to sex.

8. The sentence- “The operation went smoothly”- has changed as follows: the operation
went without difficulties.

9. Post-operative EVAR, the patient was treated with antiplatelet, lipid-lowering and blood pressure control. After open surgery, Antiplatelet, blood pressure control, and lipid control treatments continue to be maintained.